Author’s response to reviews

Title: Exploring the Parkinson patients’ perspective on home-based video recording for movement analysis: a qualitative study

Authors:

Nienke de Vries (nienke.devries-farrouh@radboudumc.nl)
Katarzyna Smilowska (kasia.smilowska@gmail.com)
Joost Hummelink (joost_hummelink@hotmail.com)
Bronislaw Abramiuc (b.abramiuc@tue.nl)
Merel Van Gilst (m.m.v.gilst@tue.nl)
Bastiaan Bloem (bas.bloem@radboudumc.nl)
Peter De With (p.h.n.de.with@tue.nl)
Sebastiaan Overeem (s.overeem@tue.nl)

Version: 4 Date: 16 Nov 2018

Author’s response to reviews:

Dear Dr. Chen,

Thank you very much for reviewing our manuscript entitled "Exploring the Parkinson patients’ perspective on home-based video recording for movement analysis: a qualitative study".

We have accommodated all editorial requests in a clean, updated version of our manuscript.

We hope that our revised manuscript can now be considered for publication in BMC Neurology.

We look forward to receiving your reaction!

With best wishes,

Nienke de Vries
1. Include ‘consent for publication’ in the declarations section

We have included the following:

Consent for publication

Not applicable

2. Include ‘availability of data and material’ in the declarations section

We have included the following:

Availability of data and material

The datasets generated and analysed during the current study are not publicly available due to its qualitative nature which makes it possible to identify individual patients. Data are available from the corresponding author on reasonable request.

3. Specify the individual contributions of all authors

We have updated this section as follows:

Author’s contributions

NMdV, SO and MvG were involved in the design of this study. NMdV coordinated and executed the study. NMdV and JH analyzed the data. KS and NMdV drafted the manuscript. All authors (NMdV, KS, JH, BA, MvG, BRB, PHNdW and SO) were involved in the interpretation of the results and critically reviewing the manuscript.